Author’s response to reviews

Title: The Application of Virtual Reality and Augmented Reality in Oral & Maxillofacial Surgery

Authors:

Ashraf Ayoub (Ashraf.Ayoub@glasgow.ac.uk)

Yeshwanth Pulijala (yesh@myscalpel.com)

Version: 4 Date: 07 Oct 2019

Author’s response to reviews:

I obtained an approval for each of the two pictures to be published in the manuscript. For the first figure, Prof. Chapman, provided the approval. For figure 2, this was taken from the PhD thesis of my co-author on this manuscript, he provided approval to use the image.

Authors’ Contributions section, I added each of the authors initials as requested

Availability of Data and Materials section has been amended

I upload a clean manuscript as suggested